

The Improbable Transmission of Trypanosoma cruzi to Human: The Missing Link in the Dynamics and Control of Chagas Disease

Pierre Nouvellet, Eric Dumonteil, Sébastien Gourbière

▶ To cite this version:

Pierre Nouvellet, Eric Dumonteil, Sébastien Gourbière. The Improbable Transmission of Trypanosoma cruzi to Human: The Missing Link in the Dynamics and Control of Chagas Disease. PLoS Neglected Tropical Diseases, 2013, 7 (11), pp.e2505. 10.1371/journal.pntd.0002505 . halsde-00905784

HAL Id: halsde-00905784 https://hal.science/halsde-00905784

Submitted on 18 Nov 2013 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

The Improbable Transmission of *Trypanosoma cruzi* to Human: The Missing Link in the Dynamics and Control of Chagas Disease

Pierre Nouvellet^{1,2,3}, Eric Dumonteil^{1,4}, Sébastien Gourbière^{2,5}*

1 Laboratorio de Parasitología, Centro de Investigaciones Regionales "Dr. Hideyo Noguchi", Universidad Autónoma de Yucatán, Mérida, Yucatán, Mexico, 2 EA 4218 UPVD 'Institut de Modélisation et d'Analyses en Géo-Environnement et Santé', Université de Perpignan Via Domitia, Perpignan, France, 3 Medical Research Council Centre for Outbreak Analysis and Modelling, Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom, 4 Department of Tropical Medicine, School of Public Health and Tropical Medicine, Tulane University, New Orleans, Louisiana, United States of America, 5 Centre for the Study of Evolution, School of Life Sciences, University of Sussex, Brighton, United Kingdom

Abstract

Chagas disease has a major impact on human health in Latin America and is becoming of global concern due to international migrations. Trypanosoma cruzi, the etiological agent of the disease, is one of the rare human parasites transmitted by the feces of its vector, as it is unable to reach the salivary gland of the insect. This stercorarian transmission is notoriously poorly understood, despite its crucial role in the ecology and evolution of the pathogen and the disease. The objective of this study was to quantify the probability of T. cruzi vectorial transmission to humans, and to use such an estimate to predict human prevalence from entomological data. We developed several models of T. cruzi transmission to estimate the probability of transmission from vector to host. Using datasets from the literature, we estimated the probability of transmission per contact with an infected triatomine to be 5.8×10^{-4} (95%CI: [2.6; 11.0]×10⁻⁴). This estimate was consistent across triatomine species, robust to variations in other parameters, and corresponded to 900-4,000 contacts per case. Our models subsequently allowed predicting human prevalence from vector abundance and infection rate in 7/10 independent datasets covering various triatomine species and epidemiological situations. This low probability of T. cruzi transmission reflected well the complex and unlikely mechanism of transmission via insect feces, and allowed predicting human prevalence from basic entomological data. Although a proof of principle study would now be valuable to validate our models' predictive ability in an even broader range of entomological and ecological settings, our quantitative estimate could allow switching the evaluation of disease risk and vector control program from purely entomological indexes to parasitological measures, as commonly done for other major vector borne diseases. This might lead to different quantitative perspectives as these indexes are well known not to be proportional one to another.

Citation: Nouvellet P, Dumonteil E, Gourbière S (2013) The Improbable Transmission of *Trypanosoma cruzi* to Human: The Missing Link in the Dynamics and Control of Chagas Disease. PLoS Negl Trop Dis 7(11): e2505. doi:10.1371/journal.pntd.0002505

Editor: Richard Reithinger, RTI International, United States of America

Received November 20, 2012; Accepted September 16, 2013; Published November 7, 2013

Copyright: © 2013 Nouvellet et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This investigation received financial support from the UNDP/World Bank/WHO Special Program for Research and Training in Tropical Diseases (TDR)/ International Development Research Center (IDRC), Project #A90276 and the European Union EU-FP7-PEOPLE-Intra-European Fellowship for career development (IEF) grant no. 253483. For part of this study, PN received funding from the European Union Seventh Framework Programme [FP7/2007–2013] under Grant Agreement no. 278433-PREDEMICS and thanks the MRC Centre for Outbreak Analysis and Modelling for support. This work was performed within the framework of the LABEX ECOFECT (ANR-11-LABX-0048) of Université de Lyon, within the program "Investissements d'Avenir" (ANR-11-IDEX-0007) operated by the French National Research Agency (ANR). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: gourbiere@univ-perp.fr

Introduction

Vector-borne diseases represent one of the biggest challenges to current and future human wellbeing. They have severe impacts on many tropical and subtropical countries, where they are responsible for $\sim 10\%$ of human deaths and contribute to impoverishment by imposing an annual burden of >50 millions of DALYs [1]. They also are an emerging threat for more developed countries as climate change and increasing international exchanges modify the geographic distributions of vectors and pathogens [2].

Vectorial transmission is traditionally thought to critically depend on the incubation period and the survival rate of the pathogen in the vector, and on the frequency of vector feeding on humans. This is well reflected in classical measures of transmission such as the vectorial capacity, entomological inoculation rate or the basic reproductive number, which are central to empirical and theoretical literature on the ecology, evolution and control of vector-borne diseases [3,4]. Also appearing in these standard measures is the parasite transmission efficacy from infected vectors to hosts, whose effects on vector-borne diseases has received less attention, and has frequently been assumed to be systematic, in particular for malaria transmission [5]. However, efficacy of vector transmission may become a key parameter when it takes on unusually low values, as even small variations could then have major effects on disease dynamics and the resulting prevalence in hosts [6].

The vast majority of causal agents of human vector-borne diseases, such as *Plasmodium*, *Leishmania*, dengue and other flaviviruses, are 'salivarian' pathogens. After entering the vector

Author Summary

Chagas disease is a parasitic disease affecting about 10 million people, often living in poor conditions, and the disease contributes to impede their development. As several other infectious diseases (malaria, dengue or sleeping sickness), it is transmitted by blood-feeding insect vectors. While most other human pathogens are directly injected with the vector's saliva, Chagas disease parasite is transmitted through the insect's feces that are deposited on the skin during bloodmeals, which seems to be a very inefficient process. The probability of such transmission to human has thus been very hard to estimate, although it is crucial to predict where people are at risk and design effective control strategies. Using mathematical models integrating data on vectors and humans collected across Latin America, we estimated that for several vector species transmission occurs in 1 over 900-4000 contacts with infected insects. We further showed that our estimate allows prediction of human infection rates in various ecological conditions. These models will provide health policy makers with improved indexes to better prioritize/ evaluate of the outcomes of vector control programs.

during a blood meal, the pathogens multiply inside the gut or haemolymph before spreading to the salivary glands to be directly injected to a human or a reservoir host during a subsequent blood meal. The probability of transmission from vector to a given host species is a complex process that depends on the size of the inoculate and on the within-host dynamics following inoculation. For some salivarian pathogens, the number of pathogens injected at the biting site can be measured [7], as well as the subsequent dynamics of the host-pathogen interactions [8]. Quantitative assessments of the overall resulting probability of transmission based on experimental infections gave values of 0.5 to ~ 1 per bite for *Plasmodium* spp. and dengue virus, 0.3–0.6 for African trypanosome, 0.2–0.4 for *Leishmania* spp., and as low as 0.01– 0.04 for the Japanese encephalitis virus [6,9].

There also are pathogens for which this probability of transmission can be much lower as they are unable to reach the salivary glands of the vector. The so-called 'stercorarian' transmission, sometimes considered as the ancestor of salivarian transmission [10], occurs through the contact of vector's feces and the biting wound (or a mucosa). Successful transmission requires an extraordinary combination of somewhat unlikely events. An infected vector has to defecate sufficiently close to the biting site whilst or shortly after feeding, the infected feces must be brought to the bite wound by the host by scratching, and the pathogen then has to cross the skin of the host to initiate infection [11].

Trypanosoma cruzi is one of the rare parasites that has managed to establish an endemic human infection through this transmission route, and despite its presumably low probability of transmission from vector to human, it has become a major public health problem. It is indeed the etiological agent of American trypanosomiasis, also called Chagas disease, a widely distributed neglected tropical disease in Latin America, with an estimated 8–9 million infected persons and another 25–90 million at risk of infection [12]. Although maternal and oral transmissions have been documented [13], vectorial transmission remains the main cause of human infection. This protozoan kinetoplastid parasite is transmitted by a large diversity of hematophagous bugs of the Reduvidae family to multiple species of sylvatic and domestic mammalian hosts, and at least 20 species of triatomines are involved in transmission to humans [14]. Nonetheless, parasite transmission through these highly diverse vector and host communities remains poorly understood, mostly because the probability of stercorarian transmission can hardly be estimated from experimental infection due to the complexity and rareness of the process. Estimates thus rely on indirect approaches based on a combination of entomological and epidemiological studies at fine temporal and spatial scales. Given the difficulty to collect such integrative datasets, there are currently only three estimates of the probability of stercorarian transmission of T. cruzi to its hosts. This probability was found to be 9×10^{-4} per contact with an infected vector for transmission to human [11], 2×10^{-4} to guinea pigs, a typical domestic host in many Latin American regions [15], and 6×10^{-2} to opossums, the likely ancestral mammalian host of T. cruzi [16]. Although these point estimates seem rather consistently low, their usefulness remains limited as there is no information on their confidence intervals, and no sensitivity analysis to uncertainties in the entomological and epidemiological raw data has been performed. A significant benefit of gaining a robust estimate of the probability of stercorarian transmission would be to establish a clear link between basic entomological data and the prevalence of T. cruzi infection in humans. Potentially, this could allow evaluating disease risk and vector control program in terms of parasitological rather than entomological indexes, as commonly done for other human vector borne diseases such as malaria [4] or dengue [17].

In this contribution we thus aimed at (i) providing a robust quantitative estimate of the probability of stercorarian transmission of T. *cruzi* to humans based on information available in the literature, and at (ii) determining if this estimate allows predicting the prevalence of the infection from basic entomological data.

Materials and Methods

1. General overview

Our estimate of the probability of stercorarian transmission of T. *cruzi* to humans was derived from an indirect approach relying on a standard mathematical relationship which links the number of healthy hosts to become infected, i.e. the incidence of T. *cruzi* infection, to the average number of potentially infective contacts per individual host, and the probability of pathogen transmission per potentially infective contacts [4]. Given such a relationship, field measurements of the first two quantities allow estimating indirectly the probability of transmission per contact from infected vector to host.

We expanded this standard modelling of transmission to derive the expected distribution of the number of susceptible humans acquiring the parasite in order to obtain a maximum likelihood estimate of the probability of stercorarian transmission. We first used this framework to re-analyze the data from [11], which provides entomological (vector abundance, infection rate, feeding frequency and the proportion of blood meals on humans) and epidemiological (incidence of infection in humans) data at the household scale. Next, we adjusted our model to estimate the probability of transmission to humans (i) when entomological and/ or epidemiological data are available at the village rather than at the household scale, and (ii) when epidemiological data consist of human prevalence rather than incidence. These adjustments allowed estimating the probability of stercorarian transmission from four additional datasets. We then performed sensitivity analysis of each estimate to the uncertainties in the measurement of the entomological variables.

Finally, we aimed at testing whether our modelling framework and estimates would allow predicting the prevalence of T. *cruzi* infection from limited entomological data. We thus predicted the prevalence of infection in different human populations representing nine different epidemiological settings, based on vector abundance and its *T. cruzi* infection rate combined with our estimate of the stercorarian transmission probability, and compared these predictions to the observed human prevalence.

2. Modelling of vector to human transmission

Vectorial transmission of *T. cruzi* to humans typically takes place inside the domestic habitat, and an accurate description of transmission thus needs to focus at this scale. Considering a household j(j=1...n) with $S_{j,t}$ susceptible individuals, the probability of observing $X_{j,t}$ new cases during a finite period of time *t*, follows a binomial law characterised by the probability $r_{j,t}$ for a susceptible human from that house to become infected during that period:

$$X_{j,t} \sim B(S_{j,t}, r_{j,t}). \tag{1}$$

The probability $r_{j,t}$ can be linked to the per contact transmission probability of parasites from an infected vector to a susceptible human, T, and the per human number of potentially infectious contacts with vectors during a single time unit, C_j . Formally, and as in [11], we have for the household 'j':

$$r_{j,t} = 1 - (1 - T)^{C_j t} \tag{2a}$$

which, as *T* is typically small for *T. cruzi*, can be approximated by the catalytic model [18,19];

$$r_{j,t} \cong 1 - e^{-T C_j t} = 1 - e^{-\beta_j t}$$
 (2b)

where β stands for the 'force of infection' [19] at the household scale.

The number of potentially infectious contacts per human per time unit can, in turn, be derived from the number of vectors present, N_{v_j} ; the proportion of infectious vectors, P_{v_j} ; the biting rate of vectors per time unit, b_j ; the feeding rate on humans, F_j ; and finally the total number of individuals in the household, N_{h_i} :

$$C_{j} = \frac{N_{v_{j}} P_{v_{j}} b_{j} F_{j}}{N_{h_{j}}}.$$
 (3)

Combining equations 2 and 3 leads to a non-linear relationship between the abundance of vectors and the incidence in humans.

3. Estimating the probability of transmission of *Trypanosoma cruzi* to humans

When all parameters appearing in equations 1-3 are known from field measurements at the household scale, the probability Tof transmission can be estimated together with a confidence interval using a standard maximum likelihood approach [20]. Using Eqs. 1-3 the log likelihood function to be maximized can be defined as:

$$\log(\mathcal{L}) = \sum_{j} \left[\log\left(\begin{pmatrix} S_{j,t} \\ X_{j,t} \end{pmatrix} \right) + X_{j,t} \log(r_{j,t}) + (S_{j,t} - X_{j,t}) \log(r_{j,t}) \right] (4)$$

where $\begin{pmatrix} S_{j,t} \\ X_{j,t} \end{pmatrix}$ stands for the binomial coefficient. An interval

estimate of T is then obtained drawing a maximum likelihood profile as a function of T [20].

However, in most cases, entomological and/or epidemiological data are available at the village rather than at the household scale (see Table 1). Still, a point estimate of T can be derived from Eqs. 1-3, but this requires further assumptions and calculations. Primarily, we have to assume that the parameters appearing in Eq. 3 have the same value across houses, leading to a common number of potentially infectious contacts per human $(C_i = C)$. Under such an assumption, the probability to become infected is the same in each household $(r_{i,t}=r_t)$, and the distribution of incidence for the entire village can be modelled by a unique Binomial $X_t = \sum_{i} X_{j,t} \sim B(S_t, r_t)$, where $S_t = \sum_{i} S_{j,t}$ stands for the total number of susceptible individuals in the village. The expectation of this law provides an estimate of r_t as the ratio between numbers of newly infected and susceptible individuals summed over all houses: $\hat{r}_t = X_t / S_t$. From Eq. 2b, one can derive a simple point estimate of T:

$$T = -\log(1 - \hat{r}_t)/Ct \tag{5}$$

In addition, epidemiological studies most frequently report human prevalence rather than incidence. Here again, one can use Eqs. 1– 3 to derive a point estimate of T under simple additional assumptions. Prevalence typically results from a dynamical equilibrium between the rate at which individuals become infected and the rate at which they die. Assuming that both the force of infection, β , and the human death rate, μ , are constant over time, the prevalence observed in the whole population, P_h , can be used to estimate the force of infection; $\beta = \mu P_h / (1 - P_h)$ (see Appendix S1 for details). Using equation 2b, one can then propose another simple point estimate of T;

$$T = \frac{\mu P_h}{(1 - P_h)} \cdot \frac{1}{C},\tag{6}$$

Prevalence data are sometimes provided for children under a given age A, $P_{h,a< A}$. The force of infection β can still be derived under similar assumptions, although it has to be evaluated numerically from the following equation (see Appendix S1 for calculations):

$$P_{h,a < A} = \frac{\beta}{\mu + \beta} - \frac{\mu e^{-\mu A} \left(1 - e^{-\beta A}\right)}{(\mu + \beta)(1 - e^{-\mu A})} \tag{7}$$

From Eq. 2b one can again derive a point estimate of T from the estimated value of β since

$$T = \frac{\beta}{C} \tag{8}$$

We again point out that both equations 6 or 7 are consistent with other known non-linear relationships between the abundance of vectors and the prevalence of infection in humans. Box 1 provides a guideline to incorporate the entomological and epidemiological knowledge into the modelling proposed in this contribution. Together with Appendices S1 and S2, it also summarizes the assumptions of the models and the potential limitation of this integrative approach.

Box 1: Models assumptions, applicability and associated caveats.

Case 1. When entomological and epidemiological information are available at the household scale, a likelihood estimation is possible. These data include the abundance of vectors (N_{v_j}) , the prevalence of infection in vectors (P_{v_j}) , the vector biting rate (b_j) , the feeding rate on humans (F_j) and the number of individuals in the household (N_{h_j}) . The analysis relies on equations 1–4 assuming that:

- All parameters used in the calculation of the risk per household, r_{j,t}, are determined without errors and are constant during the period over which the incidence is measured.
- The probability of transmission, T, is a constant, in particular human of different ages have the same probability of acquiring the parasite and we assumed no loss of infection because of successful treatment.
- iii) The vector's feeding rate on human does not vary with human age.

Case 2. When the above information is not available at the household scale, we used aggregated data and equation 5 which assumes, in addition to all assumptions described above, that:

iv) The number of potentially infectious contacts per human per time unit is constant across houses; $C_j = C$, and estimated from the aggregated values of each of the entomological and epidemiological parameters. This assumption is most likely wrong and would induce a bias when vector/human contacts are spatially heterogeneous. However, we estimated that this bias would not exceed a 20% underestimation in the worst case scenario (i.e. highly heterogeneous distribution of vectors, see Appendix S2 for details).

Case 3. When information on human incidence is not available, we use a catalytic model to infer incidence values from either prevalence in the total population (equation 6) or prevalence per age class (equation 7). Using this approach we assumed, in addition to all assumptions above, that:

- v) The human mortality is constant across ages.
- vi) The entomological observations made at time t must reflect the conditions in which the prevalence was measured. For instance, if using prevalence data of human under 20 years old, then entomological data are assumed to have been constant during the 20 previous years.

This last assumption is critical, and to limit potential bias, prevalence in younger groups should be used to best reflect the contemporary entomologic conditions (see table 2).

The sensitivity analyses relax the group of assumptions 1), and while informative, we stress that we would need better data not better models to improve our estimate of the probability of transmission of *T. cruzi* to human.

4. Sensitivity analysis of the estimate of the transmission probability

The assessment of the probability of transmission T using the indirect approaches described above relies on the estimates of the various quantities appearing in equations 1–3. Since they all are

subject to estimation uncertainty (Table 1), we performed a sensitivity analysis to such uncertainties in P_{v_j} , b_j , F_j and N_{v_j} by determining the distribution of T estimates that resulted from variations in the raw data.

The distributions of plausible estimates were obtained by randomly sampling 1000 values of P_{v_j} , b_j and F_j in independent zero-truncated normal distributions with mean and standard deviation estimated from the data (Table 1). When no information on the variability in measurements was reported by the authors, we used the standard deviation calculated from dataset 2, as this dataset provides the most comprehensive information about these three parameters. When species-specific information was missing, species-aggregated estimates were used. Since the average values of these parameters were typically larger than 5% (Table 1), sampling from an exact zero truncated binomial distribution would not change the results of the sensitivity analysis.

The sensitivity analysis to the density of vectors, N_{ν_i} , was performed in a slightly different manner. Since bug collection tends to underestimate vector densities, a correction factor is usually applied to estimate actual vector densities according to the efficacy of the particular collection method. Two different methods were used in these studies; M1 - timed-manual collection with insecticide spraying or aerosol to kill or dislodge triatomines (datasets 1-4), and M2 - passive surveillance by inhabitants or community participation (dataset 5). The correction factor of M1 and M2 have been previously estimated to be around 2.5 [11] and 10 [21], respectively. The sensitivity analysis of the estimate of T to variations in \mathcal{N}_{vi} was performed by varying the correction factor according to a uniform distribution within a range defined as its value ± 1.5 . This range was chosen as the maximum possible range to guarantee that samples collected with M1 reflected at least the density of bugs actually found (i.e. for the lowest correction factor to be 2.5-1.5 = 1), while keeping the mean of the correction factor equals to the value of 2.5 that was inferred from field data [11]. For consistency the same range was applied to other collection methods (see below for the definition of a last method - M3 - used for datasets that allowed evaluation of the model predictive ability). Accordingly, the range tested for the different methods (M1 : 1-4, M3 : 4.5-7.5, M2: 8.5-11.5) did not overlap, which is consistent with the general understanding that M1 is more efficient than M2, while M3 has an intermediate efficacy.

5. Predicting human prevalence

We then explored the usefulness of our models and our best estimate of the probability of transmission T (i.e. from dataset 1, see results) to predict the prevalence of infection in humans. From the literature, we selected nine additional studies (Table 2) reporting data on triatomine density and infection rates, as well as human prevalence of infection, so that predictions of human prevalence could be derived and compared with the observed values.

As for the sensitivity analyses above, uncertainties in the raw data were taken into account for the predictions of prevalence of infection in humans. We first constructed a distribution of the number of potentially infectious contacts per human (C_j) accounting for the variability in P_{v_j} , b_j , F_j and N_{v_j} , according to the same zero-truncated normal and uniform distributions as described for the estimates of T. The standard deviations of the zero-truncated normal distributions were again estimated from the data whenever possible, or we used the standard deviation calculated from the most comprehensive dataset (dataset 2). The range of the uniform distribution was determined according to the

Dataset	Species and location	Corrected vector density (N_{γ_j})	Vector infection rate (P_{γ})	Biting rate (b_j)	Feeding rate on humans (F_j)	Human per house (N_{h_j})	PIC (per year, C_j)	Incidence (yearly)	Source
-	T. <i>infestans</i> , Gran Chaco, Argentina	Per house ^{M1,a} : 287±47	Per house 37%±5	From ^c : 0.4	Per house: $46\% \pm 4$	Per house: 7±0.6	340±72	Directly in children ^e : 20.26%±8.86	[11]
2	T. infestans, South of Gran Chaco, Argentina	Per house ^{M1.a.} : 140±32	Per house: 10%±3	Per house ^c : 0.3 ± 0.04	Per house: 42%±6	Per house: 5±0.6	164±77	From prevalence in children ^d (Eq. 7–8, 2b): 2.54%	[24]
m	T. <i>infestans</i> , Gran Chaco, Argentina	Per house ^{M1,a} : 80±15	Per house: 40% \pm 5	Average from ^c : 0.4	Per house: $40\%\pm5$	Per house: 5 ± 0.5	248±56	Directly in children ^f (Eq. 5, 2b): 4.21%	[25,26]
4	T. longipennis and barberi, Jalisco, Mexico	Average per house ^{M1} : 45	Average for the village: 46%	Average from ^c : 0.2	Average for the village: 1%	Average per house in the village: 4	4	From prevalence (Eq. 6, 2b): 0.08%	[27,28]
ц	T. dimidiata, Yucatan, Mexico	Average per house ^{M2.b} : Teya: 40, Sudzal: 20, Merida: 1	Average for the village: Teya: 14%, Sudzal: 27%, Merida:48%	Average from ^c : 0.2	Average for the village: Teya: 6%, Sudzal: 6%, Merida:26%	Average per house in the village: 4	Teya: 7, Sudzal: 7, Merida: 3	From prevalence (Eq. 6, 2b): Teya: 0.26% , Sudzal: 0.68% , Merida: 0.26%	[29,30,31]
Data are p ^{M1} and ^{M2} ^a Vector de	resented as mean \pm SE (when refer to the triatomine collect instities were further corrected	n available in the original stud ion methods and the correspc for seasonality using monthly	y). Entomological data inding correction facto variations in infestatio.	were combined to est rs defined in the main n of <i>T. infestans</i> [32] o	timate the number of 1 text. r	potentially infectious c	ontact per person per	year (PIC).	

contact with infected vector.
) per
human (7
n vector to
ssion fron
<i>zi</i> transmi.
T. cruï
ty of
probabili:
timate th€
ised to es
datasets u
logical
epidemio
al and
Entomologic
Table 1.

^b for the seasonal infestation pattern of *T. dimidiata* [21]. ^cThe biting rate was estimated and corrected for seasonal variations according to [33]. Incidence was either measured directly or derived from prevalence using data on children or on all age-categories at the house or village scale. ^d children < 15 years old, ^e incidence after 2 years of exposure at the household scale, ^{fincidence} after 3 years of exposure at the village scale. doi:10.1371/journal.pntd.0002505.t001

taset Species ā	and location	Corrected vector density (N_{vj})	Vector infection rate $(P_{\eta j})$	Biting rate (<i>b</i>)	Feeding rate on humans (<i>F</i>) ^a	Human per house (N_{hj})	PIC (per year, <i>C</i>)	Source
T. brasilien	isis, T. pseudomaculata, Brazil	1.0 (D); 9.6 (P) ^{M1}	12% (D); 16% (P)	0.2	26% (D); 1% (P)	4	1.3	[34,35]
T. pallidip€	ennis, T. longipennis, Mexico	4.2 (D); 2.6 (P) ^{M3}	26%	0.2	26% (D); 1% (P)	4	5.2	[36,37]
T. infestan.	s, Peru	17.8 (D); 58.2 (P) ^{M1}	15% (D); 10% (P)	0.4	40% (D); 1% (P)	9	13.0	[38,39]
T. barberi,	T. mexicana, T. dimidiata, Mexico	5.4 (D) ^{M2}	7% (D)	0.2	26% (D)	4	1.7	[40]
T. dimidiat	te, Costa Rica	1.8 (D); 2.4 (P) ^{M1}	35% (D); 30% (P)	0.2	26% (D); 1% (P)	4	8.1	[41]
R. prolixus,	; P. geniculatus, Venezuela	10.4 (D) ^{M3}	17% (D)	0.2	Observed: 0.583	4	18.6	[42]
T. infestan.	s, Paraguay	43.5 (D) ^{M3}	20% (D)	0.4	40% (D)	9	46.3	[43]
R. prolixus,	; T. maculata, Venezuela	8.6 (D) ^{M2}	1% (D)	0.2	26% (D)	4	0.4	[44]
T. infestan.	s, Bolivia	SZ: 16.6 (D), NZ: 8.3 (D) ^{M2}	SZ: 79% (D), NZ: 37% (D)	0.4	40% (D)	7	SZ: 56.4, NZ: 13.4	[45]

was set to 0.4 in the domestic habitat [11,24] and to 0.01 in the peridomicile [27]. For other species, this south and North zone of Cochabamba, Bolivia [45]. The proportion of blood-meals on humans was selected according to the habitat of the bugs. For T. infestans, this proportion NZ: Respectively SZ. respectively. domestic and peridomestic habitat, proportion was set to 0.26 [46] and 0.01 in the doi:10.1371/journal.pntd.0002505.t002

insect collection method as described above. However, in 3 case studies (datasets 6, 11 and 12), bugs were collected using a third method; M3 - a timed-manual collection by trained personnel without insecticide spraying. We considered this method to have an intermediate efficacy, and we thus set the correction factor to 6 and varied its value within a range set to its value ± 1.5 .

We then determined the expected distribution of the number of infected individuals by sampling into the binomial distribution given by equation 1, with probabilities $r_{j,t}$ calculated from T and the distribution of C_i , and with the number of susceptible humans $S_{i,t}$ given by the number of individuals that were tested for Chagas disease in the prevalence studies. Since the expected number of infected individuals could be low, we applied a continuous form of the binomial distribution [22] and sampled it using the 'acceptance-rejection' methods [23]. This allowed determining a 95% confidence interval for the expected prevalence, and calculating a p-value as the probability for the observed prevalence to belong to the distribution of predicted prevalence.

6. Entomo-epidemiological datasets

Estimate of the probability of transmission (Datasets 1-**5).** Estimating the probability T of transmission requires detailed entomological and epidemiological data, which are only available from long-term field studies on specific biological systems. We thus re-analyzed the data presented by [11], and extensively reviewed the literature in the field of Chagas disease to identify additional systems which have been consistently investigated in the field over the last 10–15 years so that all entomological and epidemiological data required to get estimates of the probability of transmission were available (Table 1, [24-33]). These studies involve four vector species (Triatoma infestans, T. longipennis, T. barberi and T. dimidiata) from five locations in Argentina and Mexico. They described various entomological settings with corrected vector density varying from less than 10 to almost 300 triatomines per house, vector infection rate ranging from 10% to 58%, and vector feeding frequency on human of 1% to 46%. Overall, those entomological and epidemiological heterogeneities combined to produce substantial variations in the number of potentially infectious contacts that ranged from 3 to 340 contacts per human per year which, as expected, correlated with annual incidence of infection that ranged from about 0.1% to 20% (Table 1).

The first dataset provided all the entomological and epidemiological information at the house scale, allowing to calculate a log likelihood function (Eq. 4) and to derive an estimate and confidence interval for the probability of transmission T. In all the remaining datasets, some entomological (datasets 4-5) and/or epidemiological (datasets 2-5) data were only available at the village scale, so that only point estimates could be derived. When T. cruzi incidence was given in the original study (dataset 3), the estimate was calculated using Eq. 5, while when prevalence was measured in adult (datasets 4-5) or in children (2), the point estimate of T was calculated from Eq. 6 and 7–8, respectively.

Predicting human prevalence (Datasets 6-14). To evaluate whether our estimates of the probability of transmission Twould allow predicting the prevalence of infection in humans from limited entomological knowledge, we selected nine independent case studies reporting data on the average triatomine density and infection rate, and on human prevalence of infection (Table 2, [34-46]). Again, these data were obtained by careful screening of the literature. Estimates of the biting frequency and proportion of bites on human were not available in those studies, and we thus used values from dataset 1 to 5 according to the species of interest (i.e. T. infestans .vs. others) and the habitat of the vector (i.e. domicile .vs. peridomicile). These new datasets covered 7 countries from South and Central America, and 11 vector species including the 3 previous species, T. infestans, T. barberi and T. longipennis, as well as T. brasiliensis, T pseudomaculata, T maculata, T. mexicana, T. dimidiata, T. pallidipennis, T. longipennis, Rhodnius prolixus and Panstrongylus geniculatus. Vector densities varied between 1 and 46 inside the house, and between 2 and 58 in the peridomiciles. This, combined with important variation in vector infection rates ranging from 1 to 79%, led to a number of potentially infectious contacts per human and per year ranging from less than 1 to 56 which, again, correlated with the observed prevalence in humans that varied from 1.6 to 29.1%.

Results

1. Estimating the probability of *Trypanosoma cruzi* transmission to human

We first determined the probability of T. cruzi transmission using dataset 1 which includes data collected at the scale of the household [11]. A profile likelihood was drawn as a function of the probability of transmission (Figure 1). Such profile provided a maximum likelihood estimate of $T = 5.8 \times 10^{-4}$ per contact with infected bugs, which was close to the original point estimate of 9×10^{-4} obtained with the same data. The profile likelihood also provided a confidence interval around this estimate, which was $[2.6; 11.0] \times 10^{-4}$ per contact, which means that on average 900-4000 contacts with infected vector are needed for a host to become infected (Figure 1). This confidence interval was confirmed by the sensitivity analysis to variations in entomological raw data $(P_{v_i}, b_j, F_j \text{ and } N_{v_i})$. The 95% range of the sensitivity estimates was $[3.1; 14.9] \times 10^{-4}$ per contact and the corresponding distribution remained well within the confidence interval derived from the likelihood approach. Only 10% of values from the sensitivity analysis felt outside of the 95% likelihood confidence interval.

From the other four datasets, where entomological and/or epidemiological data were provided at the village scale, we derived additional point estimates of the probability of transmission of T. cruzi to human. These estimates varied from 1.5 and 1.7×10^{-4} for T. infestans in Argentina and 2.0×10^{-4} for T. longipennis and T. barberi in Mexico, to $[9.2; 3.5; 9.1] \times 10^{-4}$ for T. dimidiata in Mexico (Figure 2). They all lied within or very close to the confidence interval derived above from dataset 1 indicating that there is no major difference between the maximum likelihood estimates of T and these four estimates. The estimate for dataset 4 can be viewed as a species-averaged probability of transmission as the data do not allow to distinguish infection from either T. longipennis or T. barberi. The sensitivity analyses to variation in the entomological raw data $(P_{v_i}, b_j, F_j \text{ and } N_{v_i})$ further confirmed the high consistency of those results. The distribution of estimates obtained for the other two datasets on T. infestans were slightly less dispersed than in the first data set (Figure 2A and B) with 95% of the values ranging in $[0.7; 3.8] \times 10^{-4}$ for dataset 2, and in $[0.9; 4.5] \times 10^{-4}$ for dataset 3. The distribution obtained for T. longipennis and T. barberi (dataset 4) in Mexico were slightly broader with 95% of the estimates found within $[0.9; 6.2] \times 10^{-4}$ (Figure 2C). Finally the distributions obtained for T. dimidiata in Mexico (dataset 5), were the most variable since 95% of the values laid within $[5.0; 22.6] \times 10^{-4}$, $[1.7; 13.0] \times 10^{-4}$ and [5.1; $21.0] \times 10^{-4}$ for the villages of Teya, Sudzal and the city of Merida, respectively (Figure 2D, E and F). Nevertheless, all estimates were still very consistent with the likelihood-based confidence interval. According to our analysis, including a likelihood estimation with confidence interval, point estimations and sensitivity analyses, the probability of stercorarian transmission of *T. cruzi* to human is estimated to be in the range $10^{-4} - 2 \times 10^{-3}$ per contact.

2. Predicting the prevalence of *T. cruzi* infection from entomological data

We then attempted to predict human prevalence of infection based on the probability of transmission determined above and basic entomological data, using nine independent case-studies (Table 3). Those included *T. infestans* in Peru and in southern Cochabamba in Bolivia, *T. barberi*, *T. mexicana*, and *T. dimidiata* in Mexico and Costa Rica, *T. brasiliensis* and *T. pseudomaculata* in Brazil, *T. pallidipennis* and *T. longipennis* in Mexico, and *R. prolixus* and *P. geniculatus* in Venezuela.

In seven of these cases, our model satisfactorily predicted human prevalence (Table 3), as indicated by a lack of significant differences between observed and predicted prevalence. In three cases (T. infestans in northern Cochabamba, Bolivia, and in Paraguay, and T. maculata in Venezuela) there were statistical differences between the observed and predicted prevalence. However, in the later two cases, the observed and expected prevalences of infection were of very similar magnitude and almost included in the 95% confidence interval. Accordingly, the difference between the high level of transmission by T. infestans in Paraguay and the weak level of transmission by T. maculata in Venezuela were thus properly predicted, so that our model would not lead to any lack of appreciation of a serious health issue. Only in one instance, T. infestans in Northern Cochabamba, Bolivia, a large discrepancy was found and could not satisfactorily be explained from available data.

We thus obtained accurate predictions of human prevalence of *T. cruzi* infection over a broad range of epidemiological conditions ranging from low to high prevalence of infection (1.6-29.1 %), a wide geographic range (with 7 countries across Latin America), and 12 species of triatomines.

Discussion

The lack of a quantitative estimate of the probability of T. cruzi transmission to human through the feces of the vector has hindered the development of approaches that integrate ecological and epidemiological information on Chagas disease. These approaches have had an impressive influence in mitigating several vector-borne diseases including malaria [47], dengue [48] or leishmaniasis [49], and would help better understand the complex features of the transmission of T. cruzi and compare it with other vector-borne diseases. Based on data from the literature we built here epidemiological models to derive 6 estimates of this probability of transmission, all being of the order of 10^{-4} – 10^{-3} per contact. This primarily illustrates the paradox of Chagas disease; despite the 'milli-transmission' of the parasite from vectors to humans, the disease affects millions of people across the Americas. The quantitative knowledge of its transmission probability also opens new perspectives for the study of the disease, with key implications for both parasite evolution and public health policy.

1. The 'milli-transmission' of *Trypanosoma cruzi* to humans: Possible causes

The probability of transmission of *T. cruzi* from triatomine vectors to humans was found to be very small, 5.8×10^{-4} per contact with infected vector (95% CI [2.6; 11] × 10⁻⁴), relatively consistent across the different study systems, with point estimates



T. infestans Argentina, Gran Chaco (dataset 1)

probability of transmission

Figure 1. Maximum likelihood estimate of the probability of transmission of *T. cruzi.* (A) profile likelihood, maximum likelihood estimate (MLE) of the probability of transmission *T*, and its 95% maximum likelihood confidence interval (MLCI). (B) Distribution of the MLE of *T* obtained from the sensitivity analyses (1000 replications). Grey and black horizontal bars on the top of the figure represent the 95% MLCI (with the grey dot corresponding to the MLE) and the interval including 95% of the MLE estimates obtained from the sensitivity analysis. doi:10.1371/journal.pntd.0002505.g001



Figure 2. Sensitivity analyses of the probability of transmission of *T. cruzi.* Each panel gives the distribution of point estimates of T obtained from the sensitivity analyses (1000 replications). Panels A, B and C correspond to datasets 2, 3 and 4, respectively, while panels D, E and F correspond to each of the three villages included in dataset 5. Black bars represent the interval including 95% of the point estimates obtained from the sensitivity analysis. The grey dots and bars represent the maximum likelihood estimate (MLE) and 95% maximum likelihood confidence interval (MLCI) obtained from the dataset 1 for comparison (see Figure 1). doi:10.1371/journal.pntd.0002505.g002

Table 3. Predictions of human prevalence from basic entomological data and the probability of T. cruzi transmission.

Dataset	Species and location	Observed human prevalence	Predicted human prevalence [95% CI]	p-value
6	T. brasiliensis, T. pseudomaculata, Brazil	3.1% [1.7–4.6]	2.9% [1.7–4.4]	0.776
7	T. pallidipennis, T. longipennis, Mexico	2.0% ^a [0.01–4.8], 3.0% ^b [0.6–5.2]	2.0% ^a [1.3–3.0], 3.5% ^b [2.3–5.1]	0.999, 0.384 ^b
8	T. infestans, Peru	5.3% ^a [3.4–7.9]	6.1% ^a [4.6–7.8]	0.335ª
9	T. barberi, T. mexicana, T. dimidiata, Mexico	3.7% [1.2–6.3]	2.1% [0.1–4.3]	0.122
10	T. dimidiate, Costa Rica	11.7% [10.0–13.4]	10.2% [8.5–11.8]	0.068f
11	R. prolixus, P. geniculatus, Venezuela	15.5% [7.7–23.2]	25.0% [15.8–35.0]	0.050
12	T. infestans, Paraguay	29.1% [23.8–34.4]	36.9% [29.7–42.1]*	0.020
13	R. prolixus, T. maculata, Venezuela	1.6% [0.5–2.6]	0.7% [0.1–1.5]*	0.014
14	T. infestans, Bolivia	SZ: 24.9% ^c [22.1–27.8], NZ: 18.9% ^c [16.4–21.5]	SZ: 24.7% [22.1–27.5], NZ: 6.4% [5.2–7.6]*	SZ: 0.889, NZ:<0.001

Observed prevalences are presented together with the distribution of predicted prevalence (as described by a 95% confidence intervals) and the probability for the observed prevalence to be within the predicted distribution (*indicates a statistical difference between observation and predictions at a 95% confidence level). Prevalence of infection was measured and predicted for individuals under 15 years. ^a, under 30 years. ^b, or between 8 and 13 years. ^cof age. SZ, NZ: Respectively South and North zone of Cochabamba, Bolivia [45].

doi:10.1371/journal.pntd.0002505.t003

ranging from 1.5×10^{-4} to 9.2×10^{-4} , and in agreement with the only other point estimate available in the literature [11]. This narrow range of probabilities was observed in spite of marked differences in vector density, vector species (taxonomic, ecologic and behavioural differences), prevalence of infection in humans and vectors, resulting in very different epidemiological situations. Such a broad consistency was confirmed by our sensitivity analyses, which further supported that estimates are robust to changes in the entomological and epidemiological raw data used for their calculation.

These estimates of parasite transmission to human are similar to what has been calculated for guinea pigs [15], but differ substantially with the probability of stercorarian transmission to opossums that was estimated to be 10-100 times larger [17]. This suggests a reduced adaptation of T. cruzi to domestic hosts compared to its likely ancestral and sylvatic host, which is consistent with the much shorter period of coevolution between T. *cruzi* and humans. Indeed, estimates suggest around ~ 10000 years of coevolution of T. cruzi with humans, compared to ~ 80 millions years with the opossum [50]. Such a low probability of transmission does not mean that humans are of secondary importance or even 'dead ends' in term of parasite transmission, as suggested by the $\sim 40\%$ prevalence of infection in humans observed in 9000-years old mummies [51] as well as in today's highly endemic areas [45]. In fact, the potential amplification and dilution effects [52] that human and other domestic hosts could have on the populations of T. cruzi still remain to be properly quantified (but see [53]).

A low probability of vectorial transmission of T. *cruzi* was expected, given the succession of unlikely events required to occur and the many parameters involved. However, the narrow range of probabilities was more surprising given that all of these parameters could potentially affect parasite transmission quite dramatically. This suggests that these parameters combine in an independent way to produce an almost universal efficacy of transmission of T. *cruzi* from vectors to humans. While more accurate data may allow refining our estimate of the probability of transmission of T. *cruzi* to human, potentially detecting species specificity, the residual variations in the probability of transmission are expected to have little impact on the prevalence of infection in humans. Indeed, we were able to predict rather accurately the prevalence in humans

from infected vector density, the frequency of human blood meal, and a unique probability of vectorial transmission. Triatomine vectorial capacity is thus primarily dependent upon vector density and feeding frequencies on specific hosts, a conclusion which is consistent with the key influence of those parameters on the spread and persistence of the disease [6]. As vector demography has been documented for a variety of triatomine species and entomological context [21,53,54,55], the emerging eco-epidemiology of Chagas disease would benefit from a substantial improvement of our knowledge on vector feeding ecology. The emergence of methods based on the use of metagenomics [56] and stable isotopes, which potentially allow identifying vector trophic networks [57], should shortly allow tackling the transmission of *T. cruzi* in the context of host communities, as it has already been done successfully for the transmission of plague [58].

The very low probabilities of transmission of T. cruzi from vector to vertebrate hosts raises an obvious evolutionary question: why has T. cruzi not evolved from a stercorarian to a salivarian mode of transmission while closely related species such as T. rangeli [59] and T. brucei [60] or Leishmania [61] have been able to do so? A first line of explanation could lie in a lower 'evolvability' of T. cruzi [62]. However, there is no evidence that T. cruzi has a lower mutation rate compared to other kinetoplastids [63], and T. cruzi does experience reproduction and recombination [64] as demonstrated for other related taxa [65,66,67]. A lower 'evolvability' could also result from specific features in T. cruzi genotypes to phenotypes mapping function, which may be evaluated by mutagenesis and artificial selection experiments [62]. A second line of explanations that could explain *T. cruzi* ineffective mode of transmission are the costs associated with the migration of the parasite across the midgut, the escape of the immune response in the haemocoel, and the invasion of the salivary glands, which all may exert selective pressure to restrict the parasite to the gut. Those costs have not been identified yet, although molecular studies are progressively unravelling the interaction between T. cruzi and its vector [14,68], and insights could be gained by comparative analysis of vector-parasite interactions of the various kinetoplastids [14]. Comparisons with T. rangeli, a closely related and sympatric parasite that shares hosts and vectors with T. cruzi [69], should be especially informative as it is able to colonize the haemocoel and reach the salivary glands of its vector [59]. Finally, the selective

pressure on T. *cruzi* may be too weak given its potential for direct transmission which is known to be of evolutionary and epidemiological importance in opossum [16,70] as well as in human either because of oral or maternal transmission [13].

Our quantitative assessment of the probability of transmission of T. *cruzi* offers new opportunities to tackle these key ecoevolutionary questions, as it allows quantifying standard epidemiological measures such as R_0 or related quantities [3,4,71] which have been consistently missing in the epidemiology of Chagas disease [6, but see 72], while they are central tenets of the study of the epidemiological dynamics of malaria [4], dengue [17] and others human and livestock vector-borne diseases [73].

2. Shifting the assessment of Chagas disease risk and control programs from entomological to parasitological indexes

Our study demonstrates that the quantitative estimates of the probability of T. cruzi transmission from vector to humans allow expressing infection risk in terms of human incidence or prevalence, rather than in terms of purely entomological indexes such as vector presence/absence [74] or abundance [75]. Importantly, the entomological data used to make those predictions are basic estimates of vector abundance and infection rates that can inferred from typical entomological collections achieved by trained personnel or even low-cost studies based on community participation [76]. Although a proof of principle study is necessary to validate the proposed models' predictive ability in an even broader range of ecological settings, this approach offers a much more affordable way than large-scale serological surveys to estimate human prevalence over geographic areas and obtain better descriptions of the global and local burden of the disease. Such prevalence/incidence data would be more straightforward and explicit to interpret at all levels of public health systems for the design of epidemiological surveillance and vector control operations. In addition, the risk expressed in incidence or prevalence would likely differ from that expressed in vector presence/absence or abundance because, according to the catalytic model [18], the relationship between these variables is non-linear and follows a cumulative exponential distribution. At low vector densities the risk of human transmission increases rapidly with vector abundance, and is thus likely to be underestimated by the sole measure of vector abundance, while at large vector densities, human incidence and prevalence reach a plateau, so that variations in vector abundance have little or no influence on the already high transmission to human.

Our model can also profoundly help the assessment of the efficacy of vector control interventions, which is traditionally measured in terms of reduction in vector abundance or vector presence (infestation index) [53,77,78]. Typically, current guide-lines for vector control in several endemic countries aim at reducing triatomine presence below the somewhat arbitrary level of 5% of the houses of a community [79], based on the assumption

References

- World Health O (2010) Working to overcome the global impact of neglected tropical diseases: First WHO report on neglected tropical diseases, 2010.
- Norman FF, de Ayala AP, Perez-Molina JA, Monge-Maillo B, Zamarron P, et al. (2010) Neglected Tropical Diseases outside the Tropics. PLoS Negl Trop Dis 4: e762.
- Ross R (1910) The prevention of malaria. New York: E.P. Dutton & company. 774p.
- Smith D, McKenzie FE (2004) Statics and dynamics of malaria infection in Anopheles mosquitoes. Malar J 3: 13.
- Dye C, Williams BG (1995) Nonlinearities in the Dynamics of Indirectly-Transmitted Infections (or, does having a Vector make a Difference?); Grenfell BT, Dobson AP, editors. Cambridge: Cambridge University Press.

that this may be sufficient to dramatically reduce or even interrupt parasite transmission to humans. The modelling developed here opens the possibility to convert a reduction of vector abundance into a variation in the actual level of parasite transmission to humans, allowing to rapidly define more rational target/threshold levels of infestation for vector control. Again, one expects that for high vector densities, very large reduction in vector populations will be needed to reduce human prevalence, while at low vector abundance, small reductions in vectors could result in significant decrease in human prevalence. Nonetheless, even small residual populations of bugs due to incomplete treatment [80], development of insecticide resistance [81] or infestation by nondomiciliated vectors [21,55,82,83,84,85] will be sufficient to maintain an active transmission of T. cruzi to humans, which clearly appeals for the use of highly sensitive tools for entomological surveillance following the 'action' stage of control program [86].

In conclusion, we provided estimates of the probability of T. cruzi transmission from vector to human, which were shown to be highly consistent across vector species and epidemiological conditions. Such a new quantitative knowledge could allow expanding purely entomological indexes, which are typically calculated for triatomines, into parasitological measures (such as R₀, the so-called force of infection and resulting incidence and prevalence), as routinely done for other human vector-borne diseases such as malaria or dengue. This offers the possibility to develop a better understanding of the ecology and evolution of one of the rare stercorarian human parasites. This also is of primary interest in public health, as parasitological measures provide a more straightforward evaluation of the disease risk and a better description of the outcomes of vector control program in terms of human infection rather than vector abundance. Studies specifically designed to validate our models' predictive ability in an even broader range of entomological and ecological settings would now be worth performing to strengthen the proposed approach, and to allow for its use in a large scale operational/policy setting.

Supporting Information

Appendix S1 Method to estimate the probability of transmission when human incidence needs to be inferred from prevalence data. (PDF)

Appendix S2 Implication of assuming constant entomological condition across houses. (PDF)

Author Contributions

Conceived and designed the experiments: PN ED SG. Performed the experiments: PN ED SG. Analyzed the data: PN ED SG. Contributed reagents/materials/analysis tools: PN ED SG. Wrote the paper: PN ED SG. SG.

- Rascalou G, Pontier D, Menu F, Gourbière S (2012) Emergence and Prevalence of Human Vector-Borne Diseases in Sink Vector Populations. PLoS ONE 7: e36858.
- Kimblin N, Peters N, Debrabant A, Secundino N, Egen J, et al. (2008) Quantification of the infectious dose of *Leishmania major* transmitted to the skin by single sand flies. Proc Natl Acad Sci U S A 105: 10125–10130.
- Peters NC, Egen JG, Secundino N, Debrabant A, Kimblin N, et al. (2008) In Vivo Imaging Reveals an Essential Role for Neutrophils in Leishmaniasis Transmitted by Sand Flies. Science 321: 970–974.
- Ermert V, Fink AH, Jones AE, Morse AP (2011) Development of a new version of the Liverpool Malaria Model. I. Refining the parameter settings and mathematical formulation of basic processes based on a literature review. Malar J 10: 35.

- Goar SA (1970) *Trypanosoma-rangeli* of the stercoraria type in the process of adaptation to development of the salivaria type. Leningrad: Anon. Uspekhi Protozoologii (Progress in protozoology). 309–310 p.
- Rabinovich JE, Wisniveskycolli C, Solarz ND, Gurtler RE (1990) Probability of transmission of Chagas disease by *Triatoma infestans* (hemiptera, reduviidae) in an endemic area of Santiago-del-Estero, Argentina. Bull World Health Organ 68: 737–746.
- Hotez PJ, Bottazzi ME, Franco-Paredes C, Ault SK, Periago MR (2008) The Neglected Tropical Diseases of Latin America and the Caribbean: A Review of Disease Burden and Distribution and a Roadmap for Control and Elimination. PLoS Negl Trop Dis 2: e300.
- Prata A (2005) Clinical and epidemiological aspects of Chagas disease. Lancet Infect Dis 1: 92–100.
- Gourbière S, Dorn P, Tripet F, Dumonteil E (2012) Genetics and evolution of triatomines: from phylogeny to vector control. Heredity 108: 190–202.
- Basombrío MA, Gorla DE, Catalá S, Segura EL, Mora MC, et al. (1996) Number of Vector Bites Determining the Infection of Guinea Pigs with *Trypanosoma cruzi*. Mem Inst Oswaldo Cruz 91: 421–424.
- Rabinovich J, Schweigmann N, Yohai V, Wisnivesky-Colli C (2001) Probability of *Trypanosoma cruzi* transmission by *Triatoma infestans* (Hemiptera: Reduviidae) to the opossum *Didelphis albiventris* (Marsupialia: Didelphidae). Am J Trop Med Hyg 65: 125–130.
- Ferguson NM, Donnelly CA, Anderson RM (1999) Transmission dynamics and epidemiology of dengue: insights from age-stratified sero-prevalence surveys. Philos Trans R Soc Lond Ser B-Biol Sci 354: 757–768.
- Muench H (1959) Catalytic models in epidemiology. Cambridge, Mass.: Harvard University Press.
- Feliciangeli MD, Campbell-Lendrum D, Martinez C, Gonzalez D, Coleman P, et al. (2003) Chagas disease control in Venezuela: lessons for the Andean region and beyond. Trends Parasitol 19: 44–9.
- 20. Edwards AWF (1984) Likelihood: Cambridge University Press
- Barbu C, Dumonteil E, Gourbière S (2010) Characterization of the Dispersal of Non-Domiciliated *Triatoma dimidiata* through the Selection of Spatially Explicit Models. PLoS Negl Trop Dis 4: e777.
- Abramowitz M, Stegun I (1972) Handbook of Mathematical Functions with Formulas, Graphs and Mathematical Tables (9th printing with corrections). . New York: Wiley.
- von Neumann J, editor(1951) Various Techniques used in Connection with Random Digits. 36–38 p.
- Catalá S, Crocco LB, Morales GF (1997) *Trypanosoma cruzi* transmission risk index (TcTRI): an entomological indicator of Chagas disease vectorial transmission to humans. Acta Trop 68: 285–295.
- Gurtler RE, Cecere MC, Lauricella MA, Petersen RM, Chuit R, et al. (2005) Incidence of *Trypanosoma cruzi* infection among children following domestic reinfestation after insecticide spraying in rural North-western Argentina. Am J Trop Med Hyg 73: 95–103.
- Gurtler RE, Chuit R, Cecere MC, Castanera MB, Cohen JE, et al. (1998) Household prevalence of seropositivity for *Trypanosoma cruzi* in three rural villages in northwest Argentina: environmental, demographic, and entomologic associations. Am J Trop Med Hyg 59: 741–749.
- Bosseno M-F, Garcia LS, Baunaure F, Gastelum EM, Gutierrez MS, et al. (2006) Identification in triatomine vectors of feeding sources and *Trypanosoma* cruzi variants by heteroduplex assay and a multiplex miniexon polymerase chain reaction. Am J Trop Med Hyg 74: 303–305.
- 28. Brenière SF, Bosseno M-F, Magallón-Gastelúm E, Castillo Ruvalcaba EG, Gutierrez MS, et al. (2007) Peridomestic colonization of *Triatoma longipennis* (Hemiptera, Reduviidae) and *Triatoma barberi* (Hemiptera, Reduviidae) in a rural community with active transmission of *Trypanosoma cruzi* in Jalisco state, Mexico. Acta Trop 101: 249–257.
- Guzman-Tapia Y, Ramirez-Sierra MJ, Escobedo-Ortegon J, Dumonteil E (2005) Effect of hurricane Isidore on *Triatoma dimidiata* distribution and chagas disease transmission risk in the Yucatan peninsula of Mexico. Am J Trop Med Hyg 73: 1019–1025.
- Quintal RE, Polanco GG (1977) Feeding Preferences of *Triatoma dimidiata* maculipennis in Yucatan, Mexico. Am J Trop Med Hyg 26: 176–178.
- Ramirez-Sierra MJ, Herrera-Aguilar M, Gourbière S, Dumonteil E (2010) Patterns of house infestation dynamics by non-domiciliated *Triatoma dimidiata* reveal a spatial gradient of infestation in rural villages and potential insect manipulation by *Trypanosoma cruzi*. Trop Med Int Health 15: 77–86.
- Gorla DE (1988) Seasonal effects on control strategies of Chagas' disease vectors. Rev Argent Microbiol 20: 71–80.
- Catala de Montenegro S. [Estimation of the frequency of host-vector contact in populations of *Triatoma infestans* (Klug, 1834) under natural climatic conditions.]; 1987; La Falda, Cordoba, Argentina.
- Borges-Pereira J, Sarquis O, Zauza PL, Britto C, Lima MM (2008) Epidemiology of Chagas disease in four rural localities in Jaguaruana, State of Ceara. Seroprevalence of infection, parasitemia and clinical characteristics. Rev Soc Bras Med Trop 41: 345–351.
- Sarquis O, Borges-Pereira J, Mac Cord JR, Gomes TF, Cabello PH, et al. (2004) Epidemiology of chagas disease in Jaguaruana, ceara, Brazil. I. Presence of triatomines and index of *Trypanosoma cruzi* infection in four localities of a rural area. Mem Inst Oswaldo Cruz 99: 263–270.

- Coll-Cardenas R, Espinoza-Gomez F, Maldonado-Rodriguez A, Reyes-Lopez PA, Huerta-Viera M, et al. (2004) Active transmission of human Chagas disease in Colima Mexico. Mem Inst Oswaldo Cruz 99: 363–368.
- Espinoza-Gomez F, Maldonado-Rodriguez A, Coll-Cardenas R, Hernandez-Suarez CM, Fernandez-Salas I (2002) Presence of triatominae (Hemiptera, Reduviidae) and risk of transmission of Chagas disease in Colima, Mexico. Mem Inst Oswaldo Cruz 97: 25–30.
- Levy MZ, Bowman NM, Kawai V, Waller LA, del Carpio JGC, et al. (2006) Periurban *Trypanosoma cruzi*-infected *Triatoma infestans*, Arequipa, Peru. Emerg Infect Dis 12: 1345–1352.
- Levy MZ, Kawai V, Bowman NM, Waller LA, Cabrera L, et al. (2007) Targeted Screening Strategies to Detect *Trypanosoma cruzi* Infection in Children. PLoS Negl Trop Dis 1: e103.
- Becerril-Flores MA, Rangel-Flores E, Luis Imbert-Palafox J, Gomez-Gomez JV, Figueroa-Gutierrez AH (2007) Human infection and risk of transmission of Chagas disease in Hidalgo state, Mexico. Am J Trop Med Hyg 76: 318–323.
- Zeledon T, Solano G, Burstin L, Swartzwelder JC (1975) Epidemiological pattern of Chagas' disease in an endemic area of Costa Rica. Am J Trop Med Hyg 24: 214–225.
- 42. Feliciangeli MD, Carrasco H, Patterson JS, Suarez B, Martinez C, et al. (2004) Mixed domestic infestation by *Rhodnius prolixus* stal, 1859 and *Panstrongylus geniculatus* latreille, 1811, vector incrimination, and seroprevalence for *Trypano-soma cruzi* among inhabitants in el Guamito, Lara state, Venezuela. Am J Trop Med Hyg 71: 501–505.
- 43. Rojas de Arias A, Monzón MI, Velázquez de Saldívar G, Guillén E, Arrua T N (1984) Seroepidemiología de la enfermedad de Chagas en localidades rurales de Paraguay/Chagas' disease seroepidemiology in rural villages of Paraguay. Bol Of Sanit Panam 96: 189–197.
- 44. Rojas ME, Várquez P, Villarreal MF, Velandia C, Vergara L, et al. (2008) Estudio seroepidemiológico y entomológico sobre la enfermedad de Chagas en un área infestada por *Triatoma maculata* (Erichson 1848) en el centro-occidente de Venezuela. Cad Saude Publica 24: 2323–2333.
- Medrano-Mercado N, Ugarte-Fernandez R, Butron V, Uber-Busek S, Guerra HL, et al. (2008) Urban transmission of Chagas disease in Cochabamba, Bolivia. Mem Inst Oswaldo Cruz 103: 423–430.
- Guzman-Tapia Y, Ramirez-Sierra MJ, Dumonteil E (2007) Urban infestation by *Triatoma dimidiata* in the city of Merida, Yucatan, Mexico. Vector Borne Zoonotic Dis 7: 597–606.
- Keiser J, Singer BH, Utzinger J (2005) Reducing the burden of malaria in different eco-epidemiological settings with environmental management: a systematic review. Lancet Infect Dis 5: 695–708.
- Yang TC, Lu L, Fu GM, Zhong S, Ding GQ, et al. (2009) Epidemiology and vector efficiency during a dengue fever outbreak in Cixi, Zhejiang Province, China. J Vector Ecol 34: 148–154.
- Hartemink N, Vanwambeke SO, Heesterbeek H, Rogers D, Morley D, et al. (2011) Integrated Mapping of Establishment Risk for Emerging Vector-Borne Infections: A Case Study of Canine Leishmaniasis in Southwest France. PLoS ONE 6: e20817.
- Stevens JR, Gibson W (1999) The Molecular Evolution of Trypanosomes. Parasitol Today 15: 432–437.
- Aufderheide AC, Salo W, Madden M, Streitz J, Buikstra J, et al. (2004) A 9,000year record of Chagas' disease. Proc Natl Acad Sci U S A 101: 2034–2039.
- Keesing F, Holt RD, Ostfeld RS (2006) Effects of species diversity on disease risk. Ecol Lett 9: 485–498.
- Cohen JE, Gurtler RE (2001) Modeling household transmission of American trypanosomiasis. Science 293: 694–698.
- Rabinovich JE, Himschoot P (1990) A population-dynamics simulation-model of the main vectors of chagas-disease transmission, *Rhodnius-prolixus* and *Triatomainfestans*. Ecol Model 52: 249–266.
- Gourbière S, Dumonteil E, Rabinovich JE, Minkoue R, Menu F (2008) Demographic and dispersal constraints for domestic Infestation by nondomicilated Chagas disease vectors in the Yucatan peninsula, Mexico. Am J Trop Med Hyg 78: 133–139.
- 56. Toulza E., Blanc-Mathieu R., Gourbière S. and Piganeau G.. 2012. Environmental And Evolutionary Genomics Of Microbial Algae: Power And Challenges Of Metagenomics. In 'Genomic Insights Into the Biology of Algae' (ed. Piganeau G.). Advances in Botanical Research 64: 383–427.
- Gómez-Díaz E, Figuerola J (2010) New perspectives in tracing vector-borne interaction networks. Trends Parasitol 26: 470–476.
- Stapp P, Salkeld DJ (2009) Inferring host-parasite relationships using stable isotopes: implications for disease transmission and host specificity. Ecology 90: 3268–3273.
- Guhl F, Vallejo GA (2003) *Trypanosoma* (Herpetosoma) rangeli Tejera, 1920 An updated review. Mem Inst Oswaldo Cruz 98: 435–442.
- Haag J, O'HUigin C, Overath P (1998) The molecular phylogeny of trypanosomes: evidence for an early divergence of the Salivaria. Mol Biochem Parasitol 91: 37–49.
- Bates PA (2007) Transmission of Leishmania metacyclic promastigotes by phlebotomine sand flies. Int J Parasitol 37: 1097–1106.
- 62. Pigliucci M (2008) Is evolvability evolvable? Nat Rev Genet 9: 75-82.
- Stevens J, Rambaut A (2001) Evolutionary rate differences in trypanosomes. Infect Genet Evol 1: 143–150.

- Gaunt MW, Yeo M, Frame IA, Stothard JR, Carrasco HJ, et al. (2003) Mechanism of genetic exchange in American trypanosomes. Nature 421: 936– 939.
- Akopyants NS, Kimblin N, Secundino N, Patrick R, Peters N, et al. (2009) Demonstration of Genetic Exchange During Cyclical Development of Leishmania in the Sand Fly Vector. Science 324: 265–268.
- Gibson W, Peacock L, Ferris V, Williams K, Bailey M (2008) The use of yellow fluorescent hybrids to indicate mating in *Trypanosoma brucei*. Parasit Vectors 1: 4.
- Morrison LJ, Tweedie A, Black A, Pinchbeck GL, Christley RM, et al. (2009) Discovery of Mating in the Major African Livestock Pathogen *Trypanosoma* congolense. PLoS ONE 4: e5564.
- Noireau F, Diosque P, Jansen AM (2009) Trypanosoma cruzi: adaptation to its vectors and its hosts. Vet Res 40: 26.
- Ramirez LE, Lages-Silva E, Alvarenga-Franco F, Matos A, Vargas N, et al. (2002) High prevalence of *Trypanosoma rangeli* and *Trypanosoma cruzi* in opossums and triatomids in a formerly-endemic area of Chagas disease in Southeast Brazil. Acta Trop 84: 189–198.
- Schofield C (2000) Trypanosoma cruzi the vector-parasite paradox. Mem Inst Oswaldo Cruz 95: 535–44.
- Gourbière S., Menu F (2009) Adaptive dynamics of dormancy duration variability: evolutionary trade-off and priority effect lead to sub-optimal adaptation. Evolution 63(7): 1879–1892.
- Pelosse P, Kribs-Zaleta CM, Ginoux M, Rabinovich JE, Gourbière S, Menu F (2013) Influence of Vectors' Risk-Spreading Strategies and Environmental Stochasticity on the Epidemiology and Evolution of Vector-Borne Diseases: The Example of Chagas' Disease. PLoS ONE 8(8): e70830.
- Heffernan JM, Smith RJ, Wahl LM (2005) Perspectives on the basic reproductive ratio. J Royal Soc Interface 2: 281–293.
- Sarkar S, Strutz SE, Frank DM, Rivaldi CL, Sissel B, et al. (2010) Chagas Disease Risk in Texas. PLoS Negl Trop Dis 4: e836.
- Dumonteil E, Gourbière S (2004) Predicting *Triatoma dimidiata* abundance and infection rate: A risk map for natural transmission of chagas disease in the Yucatan peninsula of Mexico. Am J Trop Med Hyg 70: 514–519.

- Dumonteil E, Ramirez-Sierra MJ, Ferral J, Euan-Garcia M, Chavez-Nunez L (2009) Usefulness of Community Participation for the Fine Temporal Monitoring of House Infestation by Non-Domiciliated Triatomines. J Parasitol 95: 469–471.
- Barbu C, Dumonteil E, Gourbière S (2011) Evaluation of Spatially Targeted Strategies to Control Non-Domiciliated *Triatoma dimidiata* Vector of Chagas Disease. PLoS Negl Trop Dis 5: e1045.
- Barbu C, Dumonteil E, Gourbière S (2009) Optimization of Control Strategies for Non-Domiciliated *Triatoma dimidiata*, Chagas Disease Vector in the Yucatan Peninsula, Mexico. PLoS Negl Trop Dis 3: e416.
- Aiga H, Sasagawa E, Hashimoto K, Nakamura J, Zúniga C, et al. (2012) Chagas Disease: Assessing the Existence of a Threshold for Bug Infestation Rate. Am J Trop Med Hyg 86: 972–979.
- Reithinger R, Tarleton RL, Urbina JA, Kitron U, Guertler RE (2009) Eliminating Chagas disease: challenges and a roadmap. Br Med J 338: b1283.
- Lardeux F, Depickère S, Duchon S, Chavez T (2010) Insecticide resistance of *Triatoma infestans* (Hemiptera, Reduviidae) vector of Chagas disease in Bolivia. Trop Med Int Health 15: 1037–1048.
- Guhl F, Pinto N, Aguilera G (2009) Sylvatic triatominae: a new challenge in vector control transmission. Mem Inst Oswaldo Cruz 104: 71–75.
- Nouvellet P, Ramirez-Sierra MJ, Dumonteil E, Gourbiere S (2011). Effects of genetic factors and infection status on wing morphology of *Triatoma dimidiata* species complex in the Yucatan peninsula, Mexico. Infect Genet Evol, 11: 1243–1249.
- Pacheco-Tucuch FS, Ramirez-Sierra MJ, Gourbière S, Dumonteil E (2012) Public street lights increase house infestation by the Chagas disease vector *Triatoma dimidiata*. PLoS ONE 7(4): e36207.
- Dumonteil E, Nouvellet P, Rosecrans K, Ramirez-Sierra MJ, Gamboa-León R et al. (2013) Eco-bio-social determinants for house infestation by nondomiciliated *Triatoma dimidiata* in the Yucatan peninsula, Mexico. PLoS Negl Trop Dis, in press.
- Moncayo A (2003) Chagas disease: Current epidemiological trends after the interruption of vectorial and transfusional transmission in the southern cone countries. Mem Inst Oswaldo Cruz 98: 577–591.